Author's response to reviews

Title: Body mass index, blood pressure, and glucose and lipid metabolism among permanent and fixed-term workers in the manufacturing industry: a cross-sectional study

Authors:

Mariko Inoue (inoue-ph@med.teikyo-u.ac.jp)
Masahide Minami (maminami@dream.com)
Eiji Yano (eyano@med.teikyo-u.ac.jp)

Version: 4
Date: 14 February 2014

Author's response to reviews: see over
Dear Dr. Natalie Pafitis:

Please find enclosed our revised manuscript titled “Body mass index, blood pressure, and glucose and lipid metabolism among permanent and fixed-term workers in the manufacturing industry: A cross-sectional study” (MS: 1532244288939671), which we would like to resubmit for publication in *BMC Public Health* as an original article.

We are most grateful for the helpful comments to the original version of our manuscript. We have taken these comments into account and herewith submit a revised version of our paper. We are also uploaded our point-by-point responses to the editors’ and reviewer’s comments. We hope that our explanations and revisions are satisfactory and the revised version of our paper is now suitable for publication in *BMC Public Health*.

We look forward to hearing from you. Thank you very much for your consideration to our research paper.

Yours sincerely,

Mariko Inoue